Author’s response to reviews

Title: Assessment of the effects of skin microneedling as adjuvant therapy for facial melasma: a pilot study.

Authors:

Emerson Lima (emersonderma@terra.com.br)
Mariana Lima (emersonderma1@terra.com.br)
Maurício Paixão (mauricio_pp@terra.com.br)
HELIO MIOT (heliomiot@gmail.com)

Version: 4 Date: 20 Nov 2017

Author’s response to reviews:

Dear Editor.

As you have solicited:

"In the ethics and consent to participate section of the Declarations, please also provide the FULL name of the ethics committee that approved your study and the institution to which it belongs to."

We included it in the manuscript.

Regards.

The authors.